Author's response to reviews

Title: Developing Search Strategies for Clinical Practice Guidelines in SUMSearch and Google Scholar and Assessing their Retrieval Performance

Authors:

Andrea Haase (andrea.u.haase@arcor.de)
Markus Follmann (markus.follmann@iqwig.de)
Guido Skipka (guido.skipka@iqwig.de)
Hanna Kirchner (hanna.kirchner@iqwig.de)

Version: 5 Date: 24 June 2007

Author's response to reviews: see over
Dear BioMed Central Editorial Production Team,

We have made the suggested formatting changes, checked the manuscript, and have resubmitted the amended version of our manuscript “Developing search strategies for clinical practice guidelines in SUMSearch and Google Scholar and assessing their retrieval performance” (Authors: Andrea Haase, Markus Follmann, Guido Skipka and Hanna Kirchner) (MS: 2143750024125566).

Thank you for giving us the opportunity to publish in your journal.

Best regards,

Andrea Haase